Author's response to reviews

Title: Evidence for an association of HLA-DRB1*15 and DRB1*09 with leprosy and the impact of DRB1*09 on disease onset in a Chinese Han population

Authors:

Fu-Ren Zhang (zhangfuren@hotmail.com)
Hong Liu (hongyue2519@hotmail.com)
Shu-Min Chen (chensm2007@tom.com)
Chang-Yuan Wang (wangchangyuan5@hotmail.com)
Chuan-Fu Zhu (sdhlau0@yahoo.cn)
Lin Zhang (zhanghailin@yahoo.com.cn)
Tong-Sheng Chu (tongshengchu@163.com)
Dian-Chang Liu (liudianchang@yahoo.com.cn)
Xiao-Xiao Yan (yanxiaoxiao1979@163.com)
Jian-Jun Liu (liuj3@gis.a-star.edu.sg)

Version: 6 Date: 19 November 2009

Author's response to reviews: see over
November 19, 2009

Dear Editors:

Thank you very much for your appreciation of our manuscript. In the revised manuscript, we have improved our article according to your comments. As requested, we have provided point-to-point responses to your comments.

We are looking forward to hearing a favorable response from you soon.

Yours sincerely,
Fu-Ren Zhang, MD, PhD

Shandong Provincial Institute of Dermatology and Venereology, 57 Jiyan Lu, Jinan 250022, Shandong Provience, P.R. China
Email: zhangfuren@hotmail.com
Tel: 86-531-87298801
Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is \textit{First name Initial Last name} (e.g. Joe F Bloggs)

Answer: yes, we do.

Abstract: this abstract does not have the correct structure. Please reformat it to use these headings: Background, Methods, Results, Conclusions. Please update this both in the manuscript and on the submission system.

Answer: we have deleted objectives part.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Email address: please include email addresses for all the authors on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com

Answer: we have listed the email addresses for all the authors on the title page.

Word counts: please remove these from the title page.

Answer: yes, we do.

Keywords: please remove these from the abstract page.

Answer: yes, we do.

Abbreviations: please move this section from the abstract page, and place it between Conclusions and Competing interests.

Answer: yes, we do.

Materials and Methods: please rename this section Methods.

Answer: yes, we do.

Additional file: please remove this letter to the editors from the submission system.

Answer: OK.

References: the reference list should contain up to 30 authors, names for each citation. The term \textit{et al.} should not be used.

Answer: yes, we do.